Thanks for the comments
Here are the corrections I made according to the Specific comments
(2) The recommended English language editing company demands a fee that I can not afford, I am sorry. I read the manuscript and tried to check again in terms of English language, besides I checked by Grammarly, added a report.
(3) I added a running title: Running title: Autophagy and Digestive Diseases
(4) I revised the formatting of the keywords as:
Keywords: Autophagy; Digestive disease; Inflammatory bowel disease, Gastrointestinal cancer.
(5) I revised the formatting of reference numbers in the text according to the instructions. I verified the order and total number of references cited to ensure that all references in the list are cited and in a correct numeric order (I had missed a few, I corrected).
(6) I replaced the figure and its legend to the end of the manuscript as a separate page. Besides I developed the figure to 300dpi.
(8) The abstract is under 200 words now
(9) I revised the main titles and subtitles of the text